Author's response to reviews

Title: The influence of the cultural climate of the training environment on physicians' self-perception of competence and preparedness for practice.

Authors:

Jamiu O Busari (j.busari@atriummc.nl)
Eduard AA Verhagen (fdmuski@yahoo.com)
Fred D Muskiet (a.a.e.verhagen@bkk.umcg.nl)

Version: 13 Date: 18 November 2008

Author's response to reviews:

There are a few typographical errors

Curacao = Curaçao
(page 9) Progran = Program
(page 16) distribution = distribution.
(page 17) "the choice".....deleted
(page 18) others = other investigators